



Guideline

Treatment of polymyalgia rheumatica: British Society for Rheumatology guideline scope

Task Toyoda (1) 1,2, Zoe Armitstead³, Sampada Bhide⁴, Serge Engamba⁵, Emma Henderson⁶, Claire Jones⁷, Pieter MacKeith⁸, Janice Maddock⁹, Gary Reynolds (1) 10, Nicola Scrafton¹¹, Manil Subesinghe (1) 12,13, Sujith Subesinghe¹⁴, Helen Twohig (1) 15, Sarah L. Mackie (1) 1,2, Max Yates^{8,16,*}

Abstract

The last British Society for Rheumatology (BSR) guideline on PMR was published in 2009. The guideline needs to be updated to provide a summary of the current evidence for pharmacological and non-pharmacological management of adults with PMR. This guideline is aimed at health-care professionals in the UK who directly care for people with PMR, including general practitioners, rheumatologists, nurses, physiotherapists, occupational therapists, pharmacists, psychologists and other health professionals. It will also be relevant to people living with PMR and organisations that support them in the public and third sector, including charities and informal patient support groups. This guideline will be developed using the methods and processes outlined in the BSR Guidelines Protocol. Here we provide a brief summary of the scope of the guideline update in development.

Lay Summary

What does this mean for patients?

PMR is a common condition that causes pain, stiffness, fatigue and difficulty in doing everyday activities. PMR is usually treated with glucocorticoids (corticosteroids, 'steroids'). However, the side effects of treatment can cause problems for many patients. Since the publication of the last guideline for PMR, new research has been published. This guideline will provide healthcare professionals and people with PMR with the information they need to reach shared decisions with clinicians about their treatment, based on the best currently available evidence. In order to do this, we have formed a guideline working group and we will follow the BSR's protocol for creating a robust clinical guideline [1].

Keywords: polymyalgia rheumatica, management, treatment.

¹Leeds Institute of Rheumatic and Musculoskeletal Medicine, University of Leeds, Leeds, UK

²NIHR Leeds Biomedical Research Centre, Leeds Teaching Hospitals NHS Trust, Leeds, UK

³Department of Rheumatology, Leeds Teaching Hospitals NHS Trust, Leeds, UK

⁴East Elmbridge Primary Care Network, Kingston Hospital NHS Foundation Trust, Kingston upon Thames, UK

⁵Norwich Practices Health Centre, Norwich, UK

⁶Department of Rheumatology, Stockport NHS Foundation Trust, Stockport, UK

⁷Powys Teaching Health Board, Bronllys Hospital, Brecon, UK

⁸Norwich Medical School, University of East Anglia, Norwich, UK

⁹Patient Expert, PMRGCAuk, UK

¹⁰Centre for Immunology and Inflammatory Diseases, Massachusetts General Hospital, Boston, USA

¹¹Department of Rheumatology, Northumbria Healthcare NHS Foundation Trust, Newcastle upon Tyne, UK

¹²King's College London and Guy's and St Thomas' PET Centre, London, UK

¹³Department of Cancer Imaging, School of Biomedical Engineering and Imaging Sciences, King's College London, London, UK

¹⁴Department of Rheumatology, Guy's and St Thomas' NHS Foundation Trust, London, UK

¹⁵School of Medicine, Keele University, Keele, UK

¹⁶Rheumatology Department, Norfolk and Norwich University Hospitals NHS Foundation Trust, Norwich, UK

^{*}Correspondence to: Max Yates, Norwich Medical School, Bob Champion Research and Education Building, Floor 2, University of East Anglia, Norwich NR4 7UQ, UK. E-mail: m.yates@uea.ac.uk

2 Task Toyoda et al.

The guideline will be developed using the methods and processes outlined in Creating Clinical Guidelines: Our Protocol [1].

Why the guideline is needed

The current BSR guideline for PMR was published in 2009 [2]. In 2015, treatment recommendations for PMR were produced collaboratively by EULAR and ACR in 2015 [3]. Since then, new PMR clinical trial evidence has emerged and a major guideline update is needed.

The average age of diagnosis of PMR is 72 years and it is rarely diagnosed in those <50 years of age [4]. In the UK, PMR is primarily managed in primary care, with specialist referral for selected cases. However, there is a need to design better care pathways for PMR [5] that reflect the aspiration of the National Health Service (NHS) long-term plan to deliver more personalized therapeutic options and personcentred care [6]. Redesign of care pathways for PMR will need updated evidence-based treatment recommendations to determine optimal care for these patients.

In this guideline we will seek to identify evidence relating to treatment of patients with PMR; we will not cover methods of PMR diagnosis. Clinical diagnosis is a matter for the judgement of an appropriately trained and experienced clinician, supported by targeted investigations depending on the clinical presentation of the individual patient and the context and setting of care. We will not cover immune checkpoint inhibitor–associated PMR because this is a special case in which the context of treatment must take into account the imperative for treatment of the underlying neoplastic condition for which the immune checkpoint inhibitor is being given.

Key facts and figures

PMR is an inflammatory rheumatic disease characterized by musculoskeletal stiffness and pain in a proximal distribution, commonly affecting those >50 years of age. The estimated incidence of PMR in the UK is $\approx 96/100\,000$ per year over the age of 40 years; epidemiological studies of PMR from Northern European countries have generally reported a higher incidence and prevalence than studies from other countries [7, 8]. PMR affects $\approx 1\%$ of the UK population, with a predisposition towards women (lifetime prevalence of 2.4% in women vs 1.7% in men) and incidence increases with age [8, 9]. There is no conclusive evidence on the aetiology of PMR, although a combination of genetic factors and environmental triggers has been proposed to contribute to risk.

The clinical spectrum of PMR is wide but it classically manifests as bilateral aches and stiffness in the shoulders, neck and hips. Stiffness, a cardinal feature of PMR, is typically worse in the mornings and improves after periods of activity but may last all day [10]. These symptoms may cause difficulty elevating the shoulders, rising from a chair, turning over in bed or getting out of bed. The onset of symptoms may be over days, weeks or sometimes months, often accompanied by systemic symptoms such as malaise, fatigue, anorexia, weight loss and generalized arthralgia. Inflammatory markers (acute phase reactants including C-reactive protein, erythrocyte sedimentation rate and plasma viscosity) are usually elevated, but fever is less common [11]. Distal musculoskeletal features have been reported in 15–30% of people

with PMR, including peripheral arthritis, distal swelling with pitting oedema and carpal tunnel syndrome [12]. Some patients initially diagnosed as PMR are later diagnosed with RA [13]. PMR can also be complicated by the development of GCA in ≈5−10% of cases [14] and a proportion of patients with PMR in secondary care cohorts may have GCA-like abnormalities on vascular imaging without any symptoms or signs of GCA [15]. In the absence of clinical features of GCA, the significance of such imaging findings is uncertain, particularly as atherosclerosis may also have a similar appearance on vascular imaging. Conversely, PMR-like symptoms are found in up to 50% of those diagnosed with GCA. PMR and GCA have some similarities, but they also have some differences.

The diagnosis of PMR is based on symptoms, signs and laboratory markers with a directed search for other conditions that can mimic PMR, based on the clinical presentation and context of care. Where there is diagnostic doubt that cannot be resolved clinically, advanced imaging may sometimes be used [16, 17], but to date, imaging tests for PMR are predominantly used for research purposes.

Current practice

The aim of treatment is to relieve PMR symptoms and maintain symptomatic relief over time, while minimizing treatment side effects. Initial treatment is with glucocorticoids: typically, 15–25 mg prednisolone daily. In practice, 29–45% of patients with PMR do not respond completely to the initial treatment and \geq 50% experience significant steroid side effects [3, 18]. The average duration of glucocorticoid therapy is usually quoted as between 1 and 2 years, but 25% of patients require >4 years of therapy [8, 19, 20]. The observed management heterogeneity of PMR arises from variations in clinical practice as well as the clinical heterogeneity of people with PMR.

It has been proposed that cumulative glucocorticoid burden might be reduced by administering glucocorticoids via periodic intramuscular or (peri)articular injection rather than via the oral route, but there are practical challenges to this as these injections must be delivered by a trained healthcare professional.

The disease course of PMR is complicated by relapses, with an estimated rate of 43% at 1 year [19]. Relapses are managed by increasing the glucocorticoid dose to the pre-relapse dose and tapering back down to the relapse dose over 4–8 weeks.

DMARDs are used successfully as a steroid-sparing agent with high efficacy in the management of many rheumatic conditions. Based on clinical trials data, the 2015 EULAR/ACR recommendations conditionally recommend early initiation of methotrexate, particularly in those at high risk of relapse and/or requiring prolonged glucocorticoid therapy [3]. However, there was no high-quality evidence to predict at the point of diagnosis which people with PMR were likely to relapse [3].

The evidence base for monitoring and follow-up for people with PMR is lacking. The current recommendations are consensus-based and guided by expert opinion. Some guidelines suggest that follow-up frequency could be as frequent as 1–4 weeks until disease remission [24], while other guidelines suggest every 1–4 months in the first year of diagnosis [2, 4]. The actual patterns of healthcare utilization of people

diagnosed with PMR have been little studied, and these data are needed in order to plan better care pathways for this group.

Regarding future pharmacological therapies, the IL-6 pathway inhibitors tocilizumab and sarilumab have been studied in new-onset and refractory PMR [21–23]. Sarilumab has now been approved by the US Food and Drug Administration for treatment of people with PMR with an inadequate response to glucocorticoids or those who cannot tolerate a glucocorticoid taper. At the time of writing, neither IL-6 pathway inhibitor is approved in the UK or Europe for treatment of PMR. Other biologic DMARDs (bDMARDs) and targeted synthetic DMARDs (tsDMARDs) have also been evaluated in phase 2 trials and some phase 3 trials are now under way.

Non-pharmacological interventions such as physiotherapy, diet and nutritional supplements and complementary therapies have been little researched in PMR. The EULAR/ACR consensus recommends personalized exercise programs to maintain muscle mass and function and reduce the risk of falls, although acknowledging the limited evidence [3]. A UK cohort study of people with PMR found only 17% were offered physiotherapy, contrasting with other musculoskeletal conditions such as adhesive capsulitis and rotator cuff tears, which had referral rates >70% [25]. Physiotherapy is widely recognized as being useful for many musculoskeletal conditions. More research is needed to understand how non-medical health professionals can best add value in managing people living with PMR.

Who the guideline is for

This guideline is for general practitioners, rheumatologists and general medicine physicians; specialist nurses and allied health professionals involved in the management of people with PMR; people with PMR and other stakeholders such as patient organizations.

There are no known equality considerations.

What the guideline will and will not cover

The group that will be covered is people with PMR.

Areas that will not be covered include diagnosis of PMR [2], GCA [26] and immune checkpoint inhibitor-induced PMR [27].

Settings that will be covered include primary care and community settings and secondary and tertiary care settings.

Activities, services or aspects of care

We will look at evidence in the following areas when developing the guideline, but it may not be possible to make recommendations in all the areas: pharmacological interventions; nonpharmacological interventions; management of relapses; followup and monitoring, including stopping treatment; outcome measures and goals for PMR treatment and patient information and support.

Previous guidance

Previous guidance includes the 2015 recommendations for the management of polymyalgia rheumatica: a EULAR/ACR collaborative initiative [4] and the 2009 BSR and BHPR guidelines for the management of polymyalgia rheumatica [2].

Key issues and draft questions

While writing this scope, we identified the following key issues and draft questions related to them. The key issues and draft questions will be framed in the Population, Intervention, Comparator, Outcome (PICO) format and used to develop more detailed review questions, which will guide the systematic review of the literature.

Glucocorticoid dose

- 1) In people with PMR (P), what is the effect of the starting dose of glucocorticoids (I/C) on short-term remission of symptoms at 2–4 weeks (O)?
- 2) In people with PMR (P), what is the effect of the starting dose of glucocorticoids (I/C) on relapse risk, cumulative glucocorticoid dose, treatment-related adverse effects, quality of life and patient experience (O)?

Glucocorticoid tapering

- 1) In people with PMR (P), what is the effect of the dose and interval of glucocorticoid tapering (I/C) on remission, relapse risk, cumulative glucocorticoid dose, treatment-related adverse effects, quality of life and patient experience (O)?
- 2) In people with PMR (P), what is the effect of prescribing a predefined glucocorticoid taper (I) on relapse risk, cumulative glucocorticoid dose, treatment-related adverse effects, quality of life and patient experience (O) compared with standard care (C)?
- 3) In people with PMR (P), what is the effect of a treat-to-target approach to treatment adjustments (I) on relapse risk, cumulative glucocorticoid dose, treatment-related adverse effects, quality of life and patient experience (O) compared with standard care (C)?
- 4) In people with PMR in clinical remission (P), what is the effect of the dose and interval of glucocorticoid tapering (I/C) on remission, relapse risk, cumulative glucocorticoid dose, treatment-related adverse effects, quality of life and patient experience (O)?

DMARDs

- In people with PMR (P), what is the effect of glucocorticoids combined with conventional synthetic DMARDs (csDMARDs) (I) on relapse risk, cumulative glucocorticoid dose, treatment-related adverse effects, quality of life and patient experience (O) compared with glucocorticoids alone (C)?
- 2) In people with PMR (P), what is the effect of glucocorticoids combined with bDMARDs or tsDMARDs (I) on relapse risk, cumulative glucocorticoid dose, treatment-related adverse effects, quality of life and patient experience (O) compared with glucocorticoids alone (C)?
- 3) In people with PMR (P), what is the effect of csDMARDs (I) on relapse risk, cumulative glucocorticoid dose, treatment-related adverse effects, quality of life and patient experience (O) compared with glucocorticoids alone (C)?
- 4) In people with PMR (P), what is the effect of bDMARDs or tsDMARDs (I) on relapse risk, cumulative glucocorticoid dose, treatment-related adverse

4 Task Toyoda et al.

effects, quality of life and patient experience (O) compared with glucocorticoids alone (C)?

- 5) In people with PMR (P), what is the effect of bDMARDs or tsDMARDs with or without glucocorticoids (I) on relapse risk, cumulative glucocorticoid dose, treatment-related adverse effects, quality of life and patient experience (O) compared with csDMARDs with or without glucocorticoids (C)?
- 6) In people with PMR (P), what is the effect of early introduction (within the first 6 months) of csDMARDs, bDMARDs or tsDMARDs (I) on relapse risk, cumulative glucocorticoid dose, treatment-related adverse effects, quality of life and patient experience (O) compared with delayed use (after 6 months) (C)?
- 7) In people with PMR who have relapsed (P), what is the effect of introduction of csDMARDs, bDMARDs or tsDMARDs (I) on relapse risk, cumulative glucocorticoid dose, treatment-related adverse effects, quality of life and patient experience (O) compared with glucocorticoid therapy alone (C)?
- 8) In people with PMR in clinical remission on csDMARDs, bDMARDs or tsDMARDs (P), with or without a stable maintenance dose of glucocorticoids, what is the effect (O) of tapering the DMARD dose (I) compared with not tapering the DMARD dose (C), while maintaining a constant dose of glucocorticoid therapy?

Managing relapses

- 1) In people with PMR who relapsed on glucocorticoid tapering (P), what is the effect of the subsequent dose and interval of glucocorticoid tapering (I/C) on remission, relapse risk, cumulative glucocorticoid dose, treatment-related adverse effects, quality of life and patient experience (O)?
- 2) In people with PMR who have relapsed on glucocorticoid tapering (P), what is the effect of increasing the glucocorticoid dose to the pre-relapse dose (I) compared with adding a csDMARD, bDMARD or tsDMARD with or without glucocorticoids (C) on remission, relapse risk, cumulative glucocorticoid dose, treatment-related adverse effects, quality of life and patient experience (O)?

Other management

- 1) In people with PMR (P), what is the effect on relapse risk, cumulative glucocorticoid dose and treatment-related adverse effects, quality of life and patient experience (O) of additional group or one-on-one care from non-medical healthcare professionals (nurse, physiotherapy, occupational therapy, psychologist) (I) compared with standard care (C)?
- 2) In people with PMR (P), what is the effect of providing written information on self-management (e.g. diet, physical activity) (I) on relapse risk, cumulative glucocorticoid dose, treatment-related adverse effects, quality of life and patient experience (O) compared with standard care (C)?

Data availability

No new data were generated or analysed in support of this research.

Authors' contributions

Task Toyoda was the fellow responsible for the initial draft. All authors have made a substantial contribution to the concept or design of the article; AND drafted the article or revised it critically for important intellectual content; AND approved the version to be published; AND agreed to be accountable for all aspects of the work in ensuring that questions related to the accuracy or integrity of any part of the work are appropriately investigated and resolved.

Funding

This work was supported by BSR. No specific funding was received from any bodies in the public, commercial or not-for-profit sectors to carry out the work described in this article. S.L.M. is supported by the Leeds Biomedical Research Centre.

Disclosure statement: S.L.M. has received consultancy fees on behalf of her institution from Roche/Chugai, Sanofi, AbbVie and AstraZeneca; was an investigator on clinical trials for Sanofi, GlaxoSmithKline and Sparrow; received speaking/lecturing fees on behalf of her institution from Roche/Chugai, Vifor UCB and Pfizer; and was a patron of the charity PMRGCAuk. No personal remuneration was received for any of the above activities. She received support from Roche/Chugai to attend EULAR2019 in person and from Pfizer to attend ACR Convergence 2021 virtually. M.Y. reports advisory board work for Galapagos and previous support from Eli Lilly, AbbVie and UCB to attend national and international conferences. All other authors have declared no conflicts of interest.

References

- British Society for Rheumatology. Creating clinical guidelines: our protocol v.5.3 Revised on behalf of SAGWG. 2022. https://www. rheumatology.org.uk/Portals/0/Documents/Guidelines/Guidelines %20Protocol%20edited%20-%20March22%20FINAL.pdf?ver= 2023-05-18-105145-707 (3 August 2023, date last accessed).
- Dasgupta B, Borg FA, Hassan N et al. BSR and BHPR guidelines for the management of polymyalgia rheumatica. Rheumatology (Oxford) 2009;49:186–90.
- Dejaco C, Singh YP, Perel P et al. 2015 recommendations for the management of polymyalgia rheumatica: a European League Against Rheumatism/American College of Rheumatology collaborative initiative. Ann Rheum Dis 2015;74:1799–807.
- Muller S, Hider SL, Helliwell T et al. Characterising those with incident polymyalgia rheumatica in primary care: results from the PMR Cohort Study. Arthritis Res Ther 2016;18:200.
- Kay L, Lanyon P, MacGregor A. Rheumatology GIRFT programme national specialty report. 2021. https://gettingitrightfirsttime.co.uk/wp-content/uploads/2021/08/Rheumatology-Jul21h-NEW.pdf (3 August 2023, date last accessed).
- National Health Service. NHS long term plan. 2019. https://www.longtermplan.nhs.uk/ (3 August 2023, date last accessed).
- Gonzalez-Gay MA, Vazquez-Rodriguez TR, Lopez-Diaz MJ et al. Epidemiology of giant cell arteritis and polymyalgia rheumatica. Arthritis Rheum 2009;61:1454–61.
- 8. Partington RJ, Muller S, Helliwell T, Mallen CD, Abdul Sultan A. Incidence, prevalence and treatment burden of polymyalgia rheumatica in the UK over two decades: a population-based study. Ann Rheum Dis 2018;77:1750–6.
- Yates M, Graham K, Watts RA, MacGregor AJ. The prevalence of giant cell arteritis and polymyalgia rheumatica in a UK primary care population. BMC Musculoskelet Disord 2016;17:285.

- Mackie SL, Hughes R, Walsh M et al. "An impediment to living life": why and how should we measure stiffness in polymyalgia rheumatica? PLoS One 2015;10:e0126758.
- 11. Michet CJ, Matteson EL. Polymyalgia rheumatica. BMJ 2008; 336:765-9.
- 12. Salvarani C, Cantini F, Macchioni P *et al.* Distal musculoskeletal manifestations in polymyalgia rheumatica: a prospective followup study. Arthritis Rheum 1998;41:1221–6.
- 13. Yates M, Kotecha J, Watts RA *et al.* Incidence of inflammatory polyarthritis in polymyalgia rheumatica: a population-based cohort study. Ann Rheum Dis 2019;78:704–5.
- Smeeth L, Cook C, Hall AJ. Incidence of diagnosed polymyalgia rheumatica and temporal arteritis in the United Kingdom, 1990– 2001. Ann Rheum Dis 2006;65:1093–8.
- 15. Hemmig AK, Gozzoli D, Werlen L *et al.* Subclinical giant cell arteritis in new onset polymyalgia rheumatica: a systematic review and meta-analysis of individual patient data. Semin Arthritis Rheum 2022;55:152017.
- Mackie SL, Koduri G, Hill CL et al. Accuracy of musculoskeletal imaging for the diagnosis of polymyalgia rheumatica: systematic review. RMD Open 2015;1:e000100.
- 17. van der Geest KSM, Treglia G, Glaudemans A *et al.* Diagnostic value of [18F]FDG-PET/CT in polymyalgia rheumatica: a systematic review and meta-analysis. Eur J Nucl Med Mol Imaging 2021;48:1876–89.
- Mazzantini M, Torre C, Miccoli M et al. Adverse events during longterm low-dose glucocorticoid treatment of polymyalgia rheumatica: a retrospective study. J Rheumatol 2012;39:552–7.
- Floris A, Piga M, Chessa E et al. Long-term glucocorticoid treatment and high relapse rate remain unresolved issues in the real-life

- management of polymyalgia rheumatica: a systematic literature review and meta-analysis. Clin Rheumatol 2022;41:19–31.
- Muller S, Hider SL, Singh Sokhal B et al. Long-term use of glucocorticoids for polymyalgia rheumatica: follow-up of the PMR Cohort Study. Rheumatol Adv Pract 2022;6:rkac034.
- 21. Bonelli M, Radner H, Kerschbaumer A *et al.* Tocilizumab in patients with new onset polymyalgia rheumatica (PMR-SPARE): a phase 2/3 randomised controlled trial. Ann Rheum Dis 2022;81:838–44.
- Spiera RF, Unizony S, Warrington KJ et al. Sarilumab for relapse of polymyalgia rheumatica during glucocorticoid taper. N Engl J Med 2023;389:1263–72.
- 23. Devauchelle-Pensec V, Carvajal-Alegria G, Dernis E *et al*. Effect of tocilizumab on disease activity in patients with active polymyalgia rheumatica receiving glucocorticoid therapy: a randomized clinical trial. JAMA 2022;328:1053–62.
- Dejaco C, Kerschbaumer A, Aletaha D et al. Treat-to-target recommendations in giant cell arteritis and polymyalgia rheumatica. Ann Rheum Dis 2024;83:48–57.
- Weddell J, Hider SL, Mallen CD, Muller S. What non-pharmacological treatments do people with polymyalgia rheumatica try: results from the PMR Cohort Study. Rheumatol Int 2022;42:285–90.
- Mackie SL, Dejaco C, Appenzeller S et al. British Society for Rheumatology guideline on diagnosis and treatment of giant cell arteritis: executive summary. Rheumatology (Oxford) 2020; 59:487–94.
- 27. Kostine M, Finckh A, Bingham CO *et al.* EULAR points to consider for the diagnosis and management of rheumatic immunerelated adverse events due to cancer immunotherapy with checkpoint inhibitors. Ann Rheum Dis 2021;80:36–48.

Are you using a treatment that addresses all 6 key manifestations of PsA?

The key clinical manifestations of PsA are joints, axial, skin, enthesitis, dactylitis and nails.1





Joint relief in PsA:

68% of patients achieved ACR50 with Cosentyx® (secukinumab) at Year 1 (observed data)2

Results from ULTIMATE (N=166). The primary endpoint of GLOESS mean change from baseline vs placebo at Week 12 was met $(-9 \text{ vs } -6, p=0.004)^{2,3}$



Skin clearance in PsO:

55% of patients achieved PASI100 at Week 52 with Cosentyx 300 mg AI (secondary endpoint, observed data, N=41)4

Results from MATURE. The co-primary endpoints PASI 75 and IGA mod 2011 0/1 at Week 12 were met for Cosentyx 300 mg (N=41) vs placebo (N=40), (95% vs 10% and 76% vs 8% respectively, p<0.0001)



Click here to visit our HCP portal and learn more



Axial joint relief in PsA:

69% of patients achieved ASAS40 at Week 52 with Cosentyx 300 mg (secondary endpoint, observed data, N=139)1

Results from MAXIMISE. The primary endpoint of ASAS20 with Cosentyx 300 mg (N=164) vs placebo (N=164) at Week 12 was met (63% vs 31% respectively, p<0.0001)1

Cosentyx is the first and only, fully human biologic that directly blocks IL-17A regardless of its source5-10



A consistent safety profile with over 8 years of real-world experience^{5,6,11}

The most frequently reported adverse reactions are upper respiratory tract infections (17.1%) (most frequently nasopharyngitis, rhinitis).⁵,

Cosentyx licensed indications in rheumatology: Cosentyx is indicated for the treatment of active psoriatic arthritis in adult patients (alone or in combination with methotrexate) when the response to previous disease-modifying anti-rheumatic drug therapy has been inadequate; active ankylosing spondylitis in adults who have responded inadequately to conventional therapy; active non-radiographic axial spondyloarthritis with objective signs of inflammation as indicated by elevated C-reactive protein and/or magnetic resonance imaging evidence in adults who have responded inadequately to non-steroidal anti-inflammatory drugs; moderate to severe plaque psoriasis in children and adolescents from the age of 6 years, and adults who are candidates for systemic therapy; active enthesitis-related arthritis in patients 6 years and older (alone or in combination with methotrexate) whose disease has responded inadequately to, or who cannot tolerate conventional therapy; active juvenile psoriatic arthritis in patients 6 years or older (alone or in combination with methotrexate) whose disease has responded inadequately to, or who cannot tolerate, conventional therapy.⁵⁶

ULTIMATE (N=166), a multicentre, randomised, double-blind, placebo-controlled, 52-week Phase III trial in patients with PsA. Patients were randomly assigned to receive either weekly subcutaneous Cosentyx (300 mg or 150 mg according to the severity of psoriasis) or placebo followed by 4-weekly dosing thereafter. The primary outcome of mean change in the ultrasound GLOESS from baseline to Week 12 was met (-9 vs -6; p=0.004).^{2,3}
MATURE (N=122), a 52-week, multicentre, double-blind, randomised, placebo-controlled, Phase III trial in patients with Ps0. Eligible patients were randomised to Cosentyx 300 mg or placebo.

MATORE (N=12), a 52-week, inditioentre, double-bound, fanournised, placebo-controlled, raise in trial in patients with PSD. Eugliste patients were Parliaminised to Cosentyx 300 mg of placebo. The co-primary endpoints were PASI75 and IGA mod 2011 0/1 response at Week 12 were met for Cosentyx 300 mg vs placebo (95% vs 10% and 76% vs 8% respectively, p<0.0001).4

MAXIMISE (N=498) a double blind, placebo-controlled, multicentre, Phase IIIb study in patients with PSA. Patients were randomised in a 1:1:1 ratio to receive Cosentyx 300 mg, 150 mg or placebo. The primary endpoint of the proportion of patients achieving and ASAS20 response with Cosentyx 300 mg at Week 12 vs placebo was met (63% vs 31% respectively, p<0.0001).1

ACR, American College of Rheumatology; AI, auto-injector; ASAS, Assessment of SpondyloArthritis International Society; BASDAI, Bath; ankylosing spondylitis disease activity index; EULAR, European Alliance of Associations for Rheumatology; GLOESS, Global EULAR and OMERACT synovitis score; IGA mod 2011 0/1, investigator global assessment modified 2011 0/1; OMERACT, outcome measures in rheumatology; PASI, psoriasis area and severity index; PsA, psoriatic arthritis; PsO, plaque psoriasis

References: 1. Baraliakos X, et al. RMD open 2019;5:e001005; 2. Conaghan PG, et al. Poster 253. Rheumatology 2022;61(Suppt1). D0I:10.1093/rheumatology/keac133.252; 3. D'Agostino MA, et al. Rheumatology 2022;61:1867–1876; 4. Sigurgeirsson B, et al. Dermatol Ther 2022;35(3):e15285; 5. Cosentyx® (seculkinumah) GB Summary of Product Characteristics; 6. Cosentyx® (seculkinumah) NI Summary of Product Characteristics; 7. Lynde CW, et al. J Am Acad Dermatol 2014;71(1):141–150; 8. Fala L. Am Health Drug Benefits 2016;9(Special Feature):60–63; 9. Schön M & Erpenbeck L. Front Immunol 2018;9:1323; 10. Gorelick J, et al. Practical Dermatol 2016;12:35–50; 11. European Medicines Agency. European public assessment report. Medicine overview. Cosentyx (secukinumab). Available at: https://www.ema.europa.eu/en/documents/overview/cosentyx-epamedicine-overview_en.pdf [Accessed May 2024].



Cosentyx® (secukinumab) Great Britain Prescribing Information.

Please refer to the Summary of Product Characteristics (SmPC) before prescribing.

Indications: Treatment of: moderate to severe plague psoriasis in adults children and adolescents from the age of 6 years who are candidates for systemic therapy; active psoriatic arthritis in adults (alone or in combination with methotrexate) who have responded inadequately to disease-modifying anti-rheumatic drug therapy; active ankylosing spondylitis in adults who have responded inadequately to conventional therapy: active non-radiographic axial spondyloarthritis (nr-axSpA) with objective signs of inflammation as indicated by elevated C-reactive protein (CRP) and/or magnetic resonance imaging (MRI) evidence in adults who have responded inadequately to non-steroidal anti-inflammatory drugs; active enthesitis-related arthritis and juvenile psoriatic arthritis in patients 6 years and older (alone or in combination with methotrexate) whose disease has responded inadequately to, or who cannot tolerate, conventional therapy; active moderate to severe hidradenitis suppurativa (acne inversa) in adults with an inadequate response to conventional systemic HS therapy. Presentations: Cosentyx 75 mg solution for injection in pre-filled syringe; Cosentyx 150 mg solution for injection in pre-filled syringe; Cosentyx 150 mg solution for injection in pre-filled pen; Cosentyx 300 mg solution for injection in pre-filled pen. Dosage & Administration: Administered by subcutaneous injection at weeks 0, 1, 2, 3 and 4, followed by monthly maintenance dosing. Consider discontinuation if no response after 16 weeks of treatment. Each 75 mg dose is given as one injection of 75 mg. Each 150 mg dose is given as one injection of 150 mg. Each 300 mg dose is given as two injections of 150 mg or one injection of 300 mg. If possible avoid areas of the skin showing psoriasis. Plaque Psoriasis: Adult recommended dose is 300 mg. Based on clinical response, a maintenance dose of 300 mg every 2 weeks may provide additional benefit for patients with a body weight of 90 kg or higher. Adolescents and children from the age of 6 years: if weight ≥ 50 kg. recommended dose is 150 mg (may be increased to 300 mg as some patients may derive additional benefit from the higher dose). If weight < 50 kg, recommended dose is 75 mg. Psoriatic Arthritis: For patients with concomitant moderate to severe plaque psoriasis see adult plaque psoriasis recommendation. For patients who are anti-TNFα inadequate responders, the recommended dose is 300 mg, 150 mg in other patients. Can be increased to 300 mg based on clinical response. Ankylosing Spondylitis: Recommended dose 150 mg. Can be increased to 300 mg based on clinical response. nr-axSpA: Recommended dose 150 mg. Enthesitis-related arthritis and juvenile psoriatic arthritis: From the age of 6 years, if weight ≥ 50 kg, recommended dose is 150 mg. If weight < 50 kg, recommended dose is 75 mg. Hidradenitis suppurativa:

Cosentyx® (secukinumab) Northern Ireland Prescribing Information.

Please refer to the Summary of Product Characteristics (SmPC) before prescribing.

Indications: Treatment of: moderate to severe plaque psoriasis in adults. children and adolescents from the age of 6 years who are candidates for systemic therapy; active psoriatic arthritis in adults (alone or in combination with methotrexate) who have responded inadequately to disease-modifying anti-rheumatic drug therapy; active ankylosing spondylitis in adults who have responded inadequately to conventional therapy; active non-radiographic axial spondyloarthritis (nr-axSpA) with objective signs of inflammation as indicated by elevated C-reactive protein (CRP) and/or magnetic resonance imaging (MRI) evidence in adults who have responded inadequately to non-steroidal anti-inflammatory drugs; active enthesitis-related arthritis and juvenile psoriatic arthritis in patients 6 years and older (alone or in combination with methotrexate) whose disease has responded inadequately to, or who cannot tolerate, conventional therapy; active moderate to severe hidradenitis suppurativa (acne inversa) in adults with an inadequate response to conventional systemic HS therapy. Presentations: Cosentyx 150 mg solution for injection in pre-filled pen; Cosentyx 300 mg solution for injection in pre-filled pen. Dosage & Administration: Administered by subcutaneous injection at weeks 0, 2, 3 and 4, followed by monthly maintenance dosing. Consider discontinuation if no response after 16 weeks of treatment. Each 150 mg dose is given as one injection of 150 mg. Each 300 mg dose is given as two injections of 150 mg or one injection of 300 mg. If possible avoid areas of the skin showing psoriasis. Plaque Psoriasis: Adult recommended dose is 300 mg monthly. Based on clinical response, a maintenance dose of 300 mg every 2 weeks may provide additional benefit for patients with a body weight of 90 kg or higher. Adolescents and children from the age of 6 years: if weight ≥ 50 kg. recommended dose is 150 mg (may be increased to 300 mg as some patients may derive additional benefit from the higher dose). If weight < 50 kg, recommended dose is 75 mg. However, 150mg solution for injection in pre-filled pen is not indicated for administration of this dose and no suitable alternative formulation is available. Psoriatic Arthritis: For patients with concomitant moderate to severe plaque psoriasis see adult plaque psoriasis recommendation. For patients who are anti-TNFa inadequate responders, the recommended dose is 300 mg, 150 mg in other patients. Can be increased to 300 mg based on clinical response. Ankylosing Spondylitis: Recommended dose 150 mg. Can be increased to 300 mg based on clinical response. nr-axSpA: Recommended dose 150 mg. Enthesitis-related arthritis and juvenile psoriatic arthritis: From the age of 6 years, if weight ≥ 50 kg, recommended dose is 150 mg. If weight < 50 kg, recommended dose

Recommended dose is 300 mg monthly. Based on clinical response, the maintenance dose can be increased to 300 mg every 2 weeks. Contraindications: Hypersensitivity to the active substance or excipients, Clinically important, active infection, Warnings & Precautions: Infections: Potential to increase risk of infections; serious infections have been observed. Caution in patients with chronic infection or history of recurrent infection. Advise patients to seek medical advice if signs/symptoms of infection occur. Monitor patients with serious infection closely and do not administer Cosentyx until the infection resolves. Non-serious mucocutaneous candida infections were more frequently reported for secukinumab in the psoriasis clinical studies. Should not be given to natients with active tuberculosis (TB) Consider anti-tuberculosis therapy before starting Cosentyx in patients with latent TB. Inflammatory bowel disease (including Crohn's disease and ulcerative colitis): New cases or exacerbations of inflammatory bowel disease have been reported with secukinumab. Secukinumab. is not recommended in patients with inflammatory bowel disease. If a natient develops signs and symptoms of inflammatory bowel disease or experiences an exacerbation of pre-existing inflammatory bowel disease, secukinumah should be discontinued and appropriate medical management should be initiated. Hypersensitivity reactions: Rare cases of anaphylactic reactions have been observed. If an anaphylactic or serious allergic reactions occur, discontinue immediately and initiate appropriate therapy. *Vaccinations*: Do not give live vaccines concurrently with Cosentyx; inactivated or non-live vaccinations may be given. Paediatric patients should receive all age appropriate immunisations before treatment with Cosentyx. Latex-Sensitive Individuals: The removable needle cap of the 75mg and 150 mg pre-filled syringe and 150mg pre-filled pen contains a derivative of natural rubber latex. Concomitant immunosuppressive therapy: Combination with immunosuppressants, including biologics, or phototherapy has not been evaluated in psoriasis studies. Cosentyx was given concomitantly with methotrexate, sulfasalazine and/or corticosteroids in arthritis studies. Caution when considering concomitant use of other immunosuppressants. Interactions: Live vaccines should not be given concurrently with secukinumab. No interaction between Cosentyx and midazolam (CYP3A4 substrate) seen in adult psoriasis study. No interaction between Cosentyx and methotrexate and/or corticosteroids seen in arthritis studies. Fertility, pregnancy and lactation: Women of childbearing potential: Use an effective method of contraception during and for at least 20 weeks after treatment. <u>Pregnancy</u>: Preferably avoid use of Cosentyx in pregnancy. Breast feeding: It is not known if secukinumab is excreted in human breast milk. A clinical decision should be made on continuation of breast feeding during Cosentyx treatment (and up to 20 weeks after discontinuation) based on benefit

of breast feeding to the child and benefit of Cosentyx therapy to the

is 75 mg. However, 150mg solution for injection in pre-filled pen is not indicated for administration of this dose and no suitable alternative formulation is available. Hidradenitis suppurativa: Recommended dose is 300 mg monthly. Based on clinical response, the maintenance dose can be increased to 300 mg every 2 weeks. Contraindications: Hypersensitivity to the active substance or excinients Clinically important, active infection. Warnings & Precautions: Infections: Potential to increase risk of infections: serious infections have been observed. Caution in patients with chronic infection or history of recurrent infection. Advise nationts to seek medical advice if signs/ symptoms of infection occur. Monitor patients with serious infection closely and do not administer Cosentyx until the infection resolves. Non-serious mucocutaneous candida infections were more frequently reported for secukinumab than placebo in the psoriasis clinical studies Should not be given to natients with active tuberculosis (TB). Consider anti-tuberculosis therapy before starting Cosentyx in patients with latent TB. Inflammatory bowel disease (including Crohn's disease and ulcerative colitis): New cases or exacerbations of inflammatory bowel disease have been reported with secukinumah. Secukinumah is not recommended in patients with inflammatory bowel disease. If a patient develops signs and symptoms of inflammatory bowel disease or experiences an exacerbation of pre-existing inflammatory bowel disease, secukinymab should be discontinued and appropriate medical management should be initiated. Hypersensitivity reactions: Rare cases of anaphylactic reactions have been observed. If an anaphylactic or serious allergic reactions occur, discontinue immediately and initiate appropriate therapy. Vaccinations: Do not give live vaccines concurrently with Cosentyx; inactivated or non-live vaccinations may be given. Paediatric patients should receive all age appropriate immunisations before treatment with Cosentyx. Latex-Sensitive Individuals: The removable needle cap of the 150mg pre-filled pen contains a derivative of natural rubber latex. Concomitant immunosuppressive therapy: Combination with immunosuppressants, including biologics, or phototherapy has not been evaluated in psoriasis studies. Cosentyx was given concomitantly with methotrexate, sulfasalazine and/or corticosteroids in arthritis studies. Caution when considering concomitant use of other immunosuppressants. Interactions: Live vaccines should not be given concurrently with secukinumab. No interaction between Cosentyx and midazolam (CYP3A4 substrate) seen in adult psoriasis study. No interaction between Cosentyx and methotrexate and/or corticosteroids seen in arthritis studies. Fertility, pregnancy and lactation: Women of childbearing potential: Use an effective method of contraception during and for at least 20 weeks after treatment. Pregnancy: Preferably avoid use of Cosentyx in pregnancy. Breast feeding: It is not known if secukinumab is excreted in human breast milk. A clinical decision should be made on

woman. Fertility: Effect on human fertility not evaluated. Adverse Reactions: Very Common (≥1/10): Upper respiratory tract infection. Common ($\geq 1/100$ to <1/10): Oral herpes, headache, rhinorrhoea, diarrhoea, nausea, fatique. Uncommon (≥1/1,000 to <1/100): Oral candidiasis, lower respiratory tract infections, neutropenia, inflammatory bowel disease. Rare (≥1/10,000 to <1/1,000): anaphylactic reactions, exfoliative dermatitis (psoriasis patients), hypersensitivity vasculitis. Not known: Mucosal and cutaneous candidiasis (including oesophageal candidiasis). Infections: Most infections were non-serious and mild to moderate upper respiratory tract infections, e.g. nasopharyngitis, and did not necessitate treatment discontinuation. There was an increase in mucosal and cutaneous (including oesophageal) candidiasis, but cases were mild or moderate in severity, non-serious, responsive to standard treatment and did not necessitate treatment discontinuation. Serious infections occurred in a small proportion of patients (0.015 serious infections reported per patient year of follow up). Neutropenia: Neutropenia was more frequent with secukinumab than placebo, but most cases were mild, transient and reversible. Rare cases of neutropenia CTCAF Grade 4 were reported. Hypersensitivity reactions: Urticaria and rare cases of anaphylactic reactions were seen. Immunogenicity: Less than 1% of patients treated with Cosentyx developed antibodies to secukinumab up to 52 weeks of treatment. Other Adverse Effects: The list of adverse events is not exhaustive, please consult the SmPC for a detailed listing of all adverse events before prescribing. Legal Category: POM. MA Number & List Price: PLGB 00101/1205 - 75 mg pre-filled syringe - £304.70; PLGB 00101/1029 - 150 mg pre-filled pen x2 £1,218.78; PLGB 00101/1030 - 150 mg pre-filled syringe x2 £1,218.78; PLGB 00101/1198 – 300 mg pre-filled pen x 1 £1218.78. PI Last Revised: June 2023. Full prescribing information, (SmPC) is available from: Novartis Pharmaceuticals UK Limited, 2nd Floor, The WestWorks Building, White City Place, 195 Wood Lane, London, W12 7FQ. Telephone: (01276) 692255.

UK | 290802 | June 2023

Adverse Event Reporting:

Adverse events should be reported. Reporting forms and information can be found at www.mhra.gov.uk/yellowcard. Adverse events should also be reported to Novartis via uk.patientsafety@novartis.com or online through the pharmacovigilance intake (PVI) tool at www.novartis.com/report.

If you have a question about the product, please contact
Medical Information on 01276 698370 or by email at
medinfo.uk@novartis.com

continuation of breast feeding during Cosentyx treatment (and up to 20 weeks after discontinuation) based on benefit of breast feeding to the child and benefit of Cosentyx therapy to the woman. Fertility: Effect on human fertility not evaluated. Adverse Reactions: Very Common (≥1/10): Upper respiratory tract infection. Common (≥1/100 to <1/10): Oral herpes, headache, rhinorrhoea, diarrhoea, nausea, fatigue. Uncommon (≥1/1,000 to <1/100): Oral candidiasis, lower respiratory tract infections, neutropenia, inflammatory bowel disease. Rare (≥1/10,000 to <1/1,000): anaphylactic reactions, exfoliative dermatitis (psoriasis patients), hypersensitivity vasculitis. Not known: Mucosal and cutaneous candidiasis (including oesophageal candidiasis). Infections: Most infections were non-serious and mild to moderate upper respiratory tract infections, e.g. nasopharyngitis, and did not necessitate treatment discontinuation. There was an increase in mucosal and cutaneous (including oesophageal) candidiasis, but cases were mild or moderate in severity, non-serious, responsive to standard treatment and did not necessitate treatment discontinuation. Serious infections occurred in a small proportion of patients (0.015 serious infections reported per patient year of follow up). Neutropenia: Neutropenia was more frequent with secukinumah than placeho, but most cases were mild, transient and reversible. Rare cases of neutropenia CTCAE Grade 4 were reported. <u>Hypersensitivity reactions.</u> Urticaria and rare cases of anaphylactic reactions were seen. Immunogenicity: Less than 1% of patients treated with Cosentyx developed antibodies to secukinumab up to 52 weeks of treatment. Other Adverse Effects: The list of adverse events is not exhaustive, please consult the SmPC for a detailed listing of all adverse events before prescribing. Legal Category: POM. MA Number & List Price: FU/1/14/980/005 150 mg pre-filled pen x2 EU/1/14/980/010 - 300 mg pre-filled pen x 1 £1218.78. Pl Last Revised: May 2023. Full prescribing information, (SmPC) is available from: Novartis Pharmaceuticals UK Limited, 2nd Floor, The WestWorks Building, White City Place, 195 Wood Lane, London, W12 7FQ. Telephone: (01276) 692255.

UK | 284832 | May 2023

Adverse Event Reporting:

Adverse events should be reported. Reporting forms and information can be found at www.mhra.gov.uk/yellowcard. Adverse events should also be reported to Novartis via uk.patientsafety@novartis.com or online through the pharmacovigilance intake (PVI) tool at www.novartis.com/report

If you have a question about the product, please contact Medical Information on 01276 698370 or by email at medinfo.uk@novartis.com